Author's response to reviews

Title: Health promotion in primary and secondary schools in Denmark: Time trends and associations with schools' and students' characteristics

Authors:

Kirsten Nabe-Nielsen (nabe@sund.ku.dk)
Rikke Krølner (rkr@si-folkesundhed.dk)
Laust H Mortensen (lamo@sund.ku.dk)
Marie B Jørgensen (mbj@nrcwe.dk)
Finn Diderichsen (fidi@sund.ku.dk)

Version: 2 Date: 18 August 2014

Author's response to reviews:

Dear Editor

I haven’t received reviewers’ comments yet.

In this revised version, we added the necessary information about Ethics, Consent statement, Competing interests, and Authors contributions.

Kind regards,
Kirsten Nabe-Nielsen